Author's response to reviews

Title: Sex differences in responses to antiretroviral treatment in South African HIV-infected children on ritonavir-boosted lopinavir- and nevirapine-based treatment

Authors:

Stephanie Shiau (ss2568@columbia.edu)
Louise Kuhn (lk24@columbia.edu)
Renate Strehlau (renate.strehlau@wits.ac.za)
Leigh Martens (leigh.martens@wits.ac.za)
Helen McIlleron (helen.mcilleron@uct.ac.za)
Sandra Meredith (sandra.meredith@uct.ac.za)
Lubbe Wiesner (lubbe.wiesner@uct.ac.za)
Ashraf Coovadia (ashraf.coovadia@wits.ac.za)
Elaine J Abrams (eja1@columbia.edu)
Stephen M Arpadi (sma2@columbia.edu)

Version: 2 Date: 17 October 2013

Author's response to reviews:

Dear Editors:

Thank you for the information on formatting.

As requested, we have

1) Added our trial registration number
2) Added a competing interests section
3) Added a conclusions section
4) Updated our figure legends

Best regards,

Stephen Arpadi